Author's response to reviews

Title: Effect of crude leaf extract of Osyris quadripartita on Plasmodium berghei in Swiss albino mice

Authors:

Senait Girma (girmasen21@gmail.com)
Mirutse Giday (mirutseg@yahoo.com)
Berhanu Erko (berhanue@yahoo.com)
Hassen Mamo (binmamo@yahoo.com)

Version: 2
Date: 26 March 2015

Author's response to reviews:

Thursday, March 26, 2015

Dear sir/madam,

We wanted to publish our manuscript in BMC Complementary and Alternative Medicine because we found it very appropriate as it publishes articles on various aspects of alternative medicine. Secondly, it has an Open Access policy which is a great advantage for the scientific community in low-income countries where the practice of traditional medicine prevails. This makes wider distribution as well as free redistribution of published articles easy. Further, the journal has a relatively faster publication schedule without compromising rigorous peer-review process and quality assurance. It has easy-to-follow instructions for authors, has no space restrictions; and allows convenient online submission process which is saved if interrupted.

We have no problems with the journal’s peer-review and editorial policies. Its modest policies encourage and attract researchers all over the world. Finally, BioMed Central provides waiver of the manuscript-processing charge required upon acceptance to authors based in low-income countries.

The corresponding author confirms that all co-authors have agreed to the submission to the journal and the manuscript is not currently under submission in any other journal.

The authors declare that there is no potential competing interest.

The following is a list of potential reviewers of the manuscript and their contact details.

1. Wote Amelo:– School of Pharmacy, Jimma University, Jimma, Ethiopia. Email: wote.amelo@juedu.et
2. Rohela Mahmud:– Department of Parasitology, Faculty of Medicine, University of Malaya, Kuala Lumpur 50603, Malaysia Email: rohela@ummc.edu.my
3. Dapper DV: Department of Human Physiology, Faculty of Basic Medical Sciences, College of Health Sciences, University of Port Harcourt, Port Harcourt, Nigeria. Email: dapperdv2001@yahoo.com

4. Sedigheh Zakeri: Malaria and Vector Research Group (MVRG), Biotechnology Research Center, Institut Pasteur of Iran, Tehran, Iran, Pasteur Avenue, P. O. Box 1316943551, Tehran, Iran. Email: zakeris@yahoo.com

Kind regards,

Hassen Mamo ((PhD)

Corresponding author